Author's response to reviews

Title: Students' perceptions about the transition to the clinical phase of a medical curriculum with preclinical patient contacts; a focus group study.

Authors:

Merijn B Godefrooij (Merijn.Godefrooij@hag.unimaas.nl)
Agnes D Diemers (A.Diemers@sk.unimaas.nl)
Albert JJA Scherpbier (A.Scherpbier@oifdg.unimaas.nl)

Version: 4 Date: 25 March 2010

Author's response to reviews:

According to your request we made the following revisions in the manuscript:

On the section Authors' contributions on p.28 we added the following sentence at the end of the section: "All authors read and approved the final manuscript."

Furthermore we removed all figures and table placement text.